Author’s response to reviews

Title: A novel compound heterozygous variant of the SLC12A3 gene in Gitelman syndrome pedigree

Authors:

Yixin Chen (chenyixin29@163.com)
Ziyi Zhang (3090101142@zju.edu.cn)
xihua Lin (haxi1986@163.com)
Qianqian Pan (sana1y@vip.qq.com)
Fenping Zheng (zhengfenpingjindi040605@aliyun.com)
Hong Li (srrshnfm@zju.edu.cn)

Version: 3 Date: 02 Jan 2018

Author’s response to reviews:

Dear Editor-in-Chief,

Thank you very much for having our manuscript (ID: MGTC-D-17-00230) entitled “A novel compound heterozygous variant of the SLC12A3 gene in Gitelman syndrome pedigree” reviewed and for giving us an opportunity to revise the manuscript. We also deeply appreciate the editors for their advice of the manuscript, based on which we have revised the manuscript.

In the “Availability of data and materials” section, we have submitted the two variants separately to ClinVar. We also have sent our application for obtaining the accession numbers for a publication before the submission is fully processed. However, we have not received a letter in reply due to the holiday season in the U.S..

Therefore, we have included all supplementary data in the “Additional files” section (supplementary table 1-3 and figure 1). Please contact us if you have any questions.

Thank you for your re-consideration of our manuscript and we are looking forward to your favorable decision.
Sincerely yours,

Hong Li

Department of Endocrinology, Sir Run Run Shaw Hospital, Zhejiang University, School of Medicine.

Address: 3 East Qingchun Road, Hangzhou, China, CN 310016.

Email: srrshnfm@zju.edu.cn

Tel: +86-13588706366